

Paleoceanography and Paleoclimatology

COMMENTARY

10.1029/2020PA004174

Key Points:

- The importance of a paper by Modestou et al. (2020) is explained, which reports middle Miocene benthic foraminifera clumped isotope data
- Ways in which a large Antarctic ice sheet can be reconciled with a warm deep ocean and moderately high CO₂ are explored
- More broadly, I argue that recent advances in proxy methodology are resulting in ever increasing confidence in paleoclimate reconstructions

Supporting Information:

• Supporting Information S1

Correspondence to:

D. Evans, evans@em.uni-frankfurt.de

Citation:

Evans, D. (2021). Deep heat: Proxies, Miocene ice, and an end in sight for paleoclimate paradoxes? *Paleoceanography and Paleoclimatology*, *36*, e2020PA004174. https://doi. org/10.1029/2020PA004174

Received 27 NOV 2020 Accepted 22 JAN 2021

© 2021. The Authors.

This is an open access article under the terms of the Creative Commons Attribution-NonCommercial-NoDerivs License, which permits use and distribution in any medium, provided the original work is properly cited, the use is non-commercial and no modifications or adaptations are made.

Deep Heat: Proxies, Miocene Ice, and an End in Sight for Paleoclimate Paradoxes?

David Evans¹ D

¹Institute of Geosciences, Goethe University Frankfurt, Frankfurt am Main, Germany

Abstract The mid Miocene represents an important target for paleoclimatic study because the atmospheric CO_2 concentration ranged from near modern values to ~800 ppm, while a large, dynamic Antarctic ice sheet was likely to have been present throughout much of this interval. In this special issue, Modestou et al. (2020) (doi.org/10.1029/2020PA003927) reconstruct deep ocean warmth based on the clumped isotopic composition of benthic foraminifera, a technique that allows the ice volume and thermal components of the benthic oxygen isotope stack to be separated. These data reveal a very warm deep ocean while simultaneously suggesting that continental ice volume may, at times, have been greater than today. Here, I review these results in the context of recent developments in geochemical proxies and ice sheet modeling, and explore how the presence of a large Miocene ice sheet could be reconciled with CO_2 at least as high as present. More broadly, I argue that many of the 'paradoxes' that pepper the paleoclimate literature result as much from our imperfect understanding of the proxies, as from our understanding of the climate system. Robust proxies with a well-understood mechanistic basis, as employed by Modestou et al. (2020), as well as advances in model-data comparability usher in a new era of palaeoclimate research; an exciting future of untangling Earth's myriad past climate states awaits.

Plain Language Summary Reconstructing climate variation in Earth's geologic past informs us of the broad features of warm climates, which is relevant to preparing for climate change over the coming centuries. Moreover, these data can be compared to state-of-the-art climate models, which provides a test of the degree to which our models can reproduce warm climate states. A paper recently published in this journal applies a new method in order to reconstruct the temperature of the deep ocean in the middle Miocene (between 17 and 12 million years ago), when the atmospheric CO_2 concentration was naturally similar to or higher than it is today. Coupled with decades of previous study, these exciting results depict an unfamiliar world characterized by a warm deep ocean, and yet a large ice sheet was present on Antarctica. Both models and data agree that the Antarctic ice sheet in the Miocene was highly responsive to changes in the atmospheric CO_2 concentration, a clear cause of concern in the context of ongoing anthropogenic climate change.

1. Introduction

The middle Miocene (~17–12 Ma) is one of the key study intervals of the Cenozoic with respect to understanding past warm climate states. The Miocene climatic optimum (MCO, ~16.5–15 Ma) is the most recent interval with atmospheric CO₂ substantially elevated above that of the early 21st century, with boron isotope and alkenone δ^{13} C-derived estimates (Sosdian et al., 2018; Super et al., 2018) constraining peak CO₂ to around 400–800 ppm, declining to 200–400 ppm after the Miocene climate transition (MCT, ~14.5–13 Ma). The atmospheric CO₂ concentration of the MCO gave rise to a profoundly different world to today, with global mean surface temperature ~3–6°C higher than preindustrial times (Hansen et al., 2013; Tierney et al., 2020) and a substantially reduced latitudinal temperature gradient (Goldner et al., 2014); also see the review paper in this issue for a more comprehensive summary of the climate and biota of the Miocene (Steinthorsdottir et al., 2020).

As is the case for most target study intervals in the Cenozoic, the oxygen isotopic composition (δ^{18} O) of deep-ocean benthic foraminifera forms the starting point and the backbone of much of our understanding of these past worlds (Westerhold et al., 2020). Benthic δ^{18} O data record some combination of sea surface temperature (SST) in the regions of deep-water formation and global ice volume, the latter of which shifts



the bulk composition of the ocean to positive values via the growth of ice sheets with a very negative isotopic composition. Building on earlier studies, a high resolution record from the equatorial Pacific (Holbourn et al., 2014) demonstrates a strongly orbitally paced climate, with peak warmth at ~15.6 Ma transitioning to peak mid Miocene glacial conditions in at least three major steps between 14.7 and ~13.8 Ma. The magnitude of the orbitally paced δ^{18} O oscillations throughout the mid-Miocene was ~ 0.5–0.75‰, lower than that of the upper Pleistocene (~1.5‰), but generally greater than the mid Pliocene (Lisiecki & Raymo, 2005). As such, benthic foraminifera δ^{18} O data (e.g. Miller et al., 2020), combined with sediment records from the ANDRILL core (Western Ross Sea), reveal a highly dynamic mid Miocene Antarctic ice sheet (Levy et al., 2016).

An enormous amount of progress has been made since the key features of climatic variation within the mid Miocene were first described (Douglas & Savin, 1975; Savin, 1977; Shackleton & Kennett, 1975; Vincent & Berger, 1985; Woodruff & Savin, 1991), although key questions remain. Foremost amongst these are, for example, the precise size of the Antarctic ice sheet and how very large ice volumes can be reconciled with intervals of overall relative warmth (see e.g. Stap et al., 2016). Lear et al. (2015) suggested that Antarctic ice volume following the MCT was likely greater than present, based on a multiproxy approach applied to benthic foraminifera. This study used methodology pioneered in earlier work (Lear et al., 2000; Rosenthal et al., 1997) to measure δ^{18} O and an independent paleothermometer (the ratio of Mg to Ca) in the same samples, in order to determine both the temperature of the deep ocean and the oxygen isotopic composition of seawater, the latter of which can be broadly related to continental ice volume. However, substantial uncertainty remained, not least because the Mg/Ca proxy is complicated by nonthermal influences such as secular changes in the Mg/Ca ratio of seawater (Evans & Müller, 2012; Horita et al., 2002) and the carbonate chemistry of the deep ocean (Lear et al., 2010).

Complex problems require new approaches such as those employed by a paper published as part of this special issue (Modestou et al., 2020). Here, I explain the importance of their data and place them into the broader context of proxy development and mid Miocene climate records, as well as providing an outlook for paleoclimate research in view of the many tools now at our disposal.

2. New Approaches, Renewed Confidence

Modestou et al. (2020) present deep sea temperature reconstructions based on the geochemistry of benthic foraminifera, but with a key difference. Rather than 'traditional' isotopic or trace metal analysis, the authors measured the clumped isotopic composition of the foraminifera shells. The basis of the proxy lies in the temperature-dependent relative abundance of heavy isotope pairs in the carbonate component of CaCO₃, in this case ¹³C bonded to ¹⁸O (e.g. Affek, 2012). The key advantage of this technique compared to previous paleothermometers is that knowledge of the isotopic or elemental composition of the seawater in which the foraminifera grew is not required, and no resolvable sensitivity to other hydrographic parameters has been found over the range that these are likely to have varied within the Phanerozoic (Tripati et al., 2015), that is, uncertainty is dominated by analytical reproducibility unlike all other paleothermometers (Figure 1). This circumvents the issue of separating temperature versus ice volume change inherent to the interpretation of traditional δ^{18} O data, or the complications associated with evolving seawater chemistry and the carbonate chemistry of the deep ocean which must be taken into account when interpreting Mg/Ca or Mg/Li measurements (Evans & Müller, 2012; Lear et al., 2010, 2015).

However, everything comes at a price. The reason that clumped isotope thermometry is applied to Miocene benthic foraminifera for the first time by Modestou et al. (2020), is that molecules containing more than one heavy isotope are extremely rare, and therefore very difficult to measure with the required precision. The first studies to report clumped isotope measurements of carbonate samples required ~15 mg of material (Ghosh et al., 2006), equivalent to >200 relatively large benthic foraminifera, which is at best challenging, and often impractical in terms of abundances in deep sea sediment cores. Later methodological advances (Meckler et al., 2014; Schmid & Bernasconi, 2010) reduced this sample size requirement by a factor of 10, opening up new possibilities in terms of carbonate archives. Modestou et al. (2020) apply this technique to samples recovered from ODP Site 761 (NW Australian margin, modern water depth 2,179 m, middle Miocene paleo water depth similar to modern (Holbourn et al., 2004)) spanning ~16.5–12 Ma, which includes



Figure 1. The approximate magnitude of the various sources of uncertainty affecting some common deep ocean temperature proxies based on the geochemistry of benthic foraminifera. Note that this assumes the worst-case scenario that no constraints on these nonthermal controls exist, which is often not the case. In addition, relative temperature reconstructions can be much more precise than this figure implies (e.g., relative changes cannot be biased if the study interval is shorter than that over which seawater chemistry can change, although the absolute temperature reconstructions may be biased if seawater chemistry was different to modern). The magnitude of the bars relating to seawater chemistry reflect both the degree and the direction which these parameters have varied in the Neogene only. The figure is intended as an approximate guide; these values may be species/site specific. A pH (CO_3^{2-}) effect on foraminifera δ^{18} O has been suggested (Zeebe, 1999) but not (yet) found in benthic species (Marchitto et al., 2014). The effect of seawater chemistry on the Mg/Li proxy is not constrained (little is known about the possible secular evolution of seawater [Li]).

much of the MCO as well as the MCT. The data show that the deep ocean cooled by \sim 3°C between the MCO and after the MCT, within uncertainty, although substantially greater than a previous estimate based on Mg/Ca (Lear et al., 2015). While this is in-line with the \sim 200–400 ppm decrease in atmospheric CO₂ across the MCT (Sosdian et al., 2018; Super et al., 2018), more surprising are the absolute temperature reconstructions; the temperature at this site during the MCO was $11.0\pm 1.7^{\circ}$ C, decreasing to $8.1\pm 1.8^{\circ}$ C after the MCT. This is 5°–8°C warmer than the water currently bathing this site, a very large change; for context, the magnitude of the temperature change in the deep ocean between the last glacial maximum (LGM) and the Holocene was \sim 3°C (Adkins et al., 2002). Encouragingly, the absolute clumped isotope temperatures are in excellent agreement with those derived from Mg/Ca data from the same site, especially when changes in bottom water carbonate chemistry are accounted for using Li/Ca (Lear et al., 2010). This adds confidence to the results overall, and more broadly to benthic Mg/Ca and Mg/Li derived temperatures throughout the Cenozoic.

Remarkable findings require careful consideration of alternative explanations. Key to the integrity of benthic clumped isotope data is that no resolvable species or group-specific calibration differences have been found (Meinicke et al., 2020). Moreover, diagenetic alteration can substantially bias both δ^{18} O and Mg/ Ca-derived reconstructions (Kozdon et al., 2013), but benthic clumped isotope data have been shown to be minimally sensitive to diagenetic recrystallization (Leutert et al., 2019) likely because this process – if present – takes place at a temperature similar to that during the life of the foraminifera (Evans et al., 2018); clumped isotope thermometry, unlike δ^{18} O, is sensitive to temperature but insensitive to the bulk isotopic composition of the sample. For these reasons, the temperature reconstructions of Modestou et al. (2020) appear robust, and do not suffer from the nonthermal influences that complicate the interpretation of traditional stable isotope and trace element proxies in terms of absolute temperature (Figure 1). Indeed, these deep ocean temperature data are supported by mid/high-latitude sea surface temperature (SST) estimates (Levy et al., 2016; Shevenell et al., 2004; Super et al., 2018), which adds support to the notion of a greatly





Figure 2. Middle Miocene reconstructions of atmospheric CO₂ (Super et al., 2018; Sosdian et al., 2018), mid latitude sea surface temperature (SST) from Mg/Ca (Shevenell et al. (2004) reinterpreted following Gray & Evans (2019)) and GDGTs (glycerol dialkyl glycerol tetraether lipids; Super et al., 2018), deep ocean temperature from clumped isotopes (Modestou et al., 2020), and seawater δ^{18} O from coupled clumped isotope- δ^{18} O measurements of benthic foraminifera. SLE denotes sea level equivalent, using the Miocene δ^{18} O-sea level relationship of Langebroek et al. (2010).

reduced latitudinal SST gradient and therefore much warmer temperatures in the regions of deep water formation (Figure 2).

3. Deep Heat but Big Ice?

The clumped isotope data come from just one site, and modeling indicates substantial heterogeneity in the thermal response of the deep ocean as a result of changes in deep water formation driven by ice growth on Antarctica (Knorr & Lohmann, 2014). Of specific relevance to the results of Modestou et al. (2020), deep water masses in the eastern Indian ocean are predominantly related to Antarctic Intermediate Water (AAIW) and Antarctic Bottom Water (AABW)/Circumpolar Deep Water (CDW), with ODP Site 761 (~2,200 m paleo water depth) currently sitting close to the depth of the boundary between AAIW and deeper water masses (Holbourn et al., 2004; Warren, 1981; Woo & Pattiaratchi, 2008). This raises the possibility that data from ODP Site 761 potentially relate to water masses that are a few degrees warmer than those bathing much of the deep ocean. However, benthic δ^{18} O at ODP Site 761 closely tracks that recorded at many other sites in other basins (e.g. Flower & Kennett, 1995; Holbourn et al., 2014) and is in close agreement with data from ODP Site 1,237 (SE Pacific; Holbourn et al., 2005) which is currently situated at 3,212 m water depth and bathed by Pacific Central Water. In addition, benthic δ^{18} O values from ODP Site 761 are in good agreement with the long term smooth through the most recent Cenozoic compilation (Westerhold et al., 2020), indicating that this site is not substantially offset from the global mean. Therefore, heterogeneity in deep ocean temperature and/or $\delta^{18}O_{sw}$ does not appear to be a complication for the interpretation of middle Miocene data from ODP Site 761. More broadly however, analysis of a multisite compilation by Cramer et al. (2009) highlighted that benthic foraminifera δ^{18} O may differ by up to ~0.8% between sites, equivalent to \sim 3°C- this is an issue that needs careful consideration when working with a data set derived from a single site.

If the clumped isotope results are indeed globally representative as suggested by the comparisons given above, the data can be coupled to tradi-

tional δ^{18} O analyses to constrain the oxygen isotopic composition of seawater through the middle Miocene, which is what Modestou et al. (2020) go on to do. This technique has been applied many times before (e.g. Elderfield et al., 2012; Lear et al., 2000), albeit with the Mg/Ca paleothermometer as the independent temperature constraint, which is influenced by several nonthermal factors (Figure 1). The new results are surprising; taken at face value, the combined clumped isotope- δ^{18} O data show that $\delta^{18}O_{sw}$ was more positive than today throughout the middle Miocene (Figure 2). Given a lack of evidence for northern hemisphere glaciation at this time, this requires an Antarctic ice sheet larger than at present, or an alternative explanation for an isotopically heavy deep ocean. One possibility is the formation of ice with an isotopic composition much lighter than at present, which could result from a topographically higher ice sheet, or a greater average distance between Antarctica and the source of precipitation. While factors such as these might be important in understanding the fine detail of the record (Langebroek et al., 2010), the magnitude of the signal is such that this clearly cannot be the principal explanation – it would require Antarctic ice with an infeasibly light isotopic composition.

Given that an isotopically heavy seawater δ^{18} O during the mid-Miocene now appears well constrained, the question is: how can atmospheric CO₂ substantially higher than preindustrial and a very warm deep ocean be reconciled with an Antarctic ice sheet larger than today? Earlier work suggested that evaporation-driven deep water formation via the southwards movement of highly saline water from the Tethys and Indian Ocean toward Antarctica could provide the mechanism of subduction of warm, salty surface waters (Wood-ruff & Savin, 1989). If correct, this would point to an unusual mode of Miocene deep water formation, rath-



er than a state-dependent relationship between surface temperature and ice volume. However, the interpretation of Woodruff and Savin (1989) has been questioned (Smart et al., 2007), while the near closure of the connection between the Tethys and Indian Ocean broadly coincident with the MCT precludes this being the only explanation because Modestou et al. (2020) reconstruct positive $\delta^{18}O_{sw}$ throughout the middle Miocene. Indeed, modeling work (Hamon et al., 2013) argues for a role of the closure of the Tethyan gateway in driving further expansion of the Antarctic ice sheet after 14 Ma, rather than providing an alternative explanation for the isotopic composition of the deep ocean.

If ocean circulation changes alone cannot solve the conundrum, the question is finally whether there is observational evidence for a large mid-Miocene Antarctic ice sheet, and if so, whether this can be reproduced by ice sheet models. Both seismic data and sedimentological evidence demonstrate that a grounded marine ice sheet was at least periodically present in the western Ross Sea, supporting the notion of a relatively large mid Miocene ice sheet (Chow & Bart, 2003; Levy et al., 2016). Moreover, by coupling a general circulation model to a regional ice sheet model, Gasson et al. (2016) were able to reconstruct the magnitude of $\delta^{18}O_{sw}$ change between the MCO and post-MCT conditions (modeled $\Delta \delta^{18}O_{sw} = 0.53\%$; Modestou et al., 2020 derive 0.6%). Importantly, this modeling effort demonstrates that the size of the ice sheet at a given atmospheric CO_2 is sensitive to Antarctic topography, as has been previously identified for other time intervals such as the Eocene Oligocene transition (Wilson et al., 2013). Specifically, Gasson et al. (2016) reconstruct an ice volume applicable to the cooler intervals of the mid Miocene that is ~35% larger than at present, driven in part by the overall higher bedrock topography and greater extent of land above sea level (marine-grounded ice sheets are more responsive to temperature change, see e.g. Pollard & Deconto, 2016). In turn, this larger than modern and isotopically lighter ice sheet would result in a positive deep ocean $\delta^{18}O_{sw}$, in line with Modestou et al. (2020). Much remains to be understood, however. It is important to note that Antarctic ice volume throughout the middle Miocene may have been much more variable than implied by the relatively low-resolution data of Modestou et al. (2020). For example, it has been suggested that peak middle Miocene warmth was potentially associated with little ice on Antarctica (Miller et al., 2020; Stap et al., 2016; discussed in more detail in Steinthorsdottir et al., 2020). In addition, the Antarctic ice sheet response to pCO₂ depends strongly on multiple factors such as the dynamics of pCO₂ change, orbital parameters, and precipitation lapse rate (Stap et al., 2019). Higher resolution proxy CO_2 and temperature datasets will help to resolve which of these factors were most important in controlling the magnitude of ice volume changes across the MCT.

Beyond the Miocene, the finding of a very warm deep ocean during the MCT has important implications to the broader field of paleoclimatology. For example, it could imply that the relationship between global mean surface temperature (GMST) and the temperature of the deep ocean is more complicated than implied by some methods that have been used to derive GMST of past warm intervals (see Hansen et al., 2013; Inglis et al., 2020). Indeed, similarly complex relationships between ice volume, deep ocean temperature, and pCO₂ have been found for other intervals (O'Brien et al., 2020). Together, these data argue for more nuanced approaches to deriving pre-Pliocene GMST from deep ocean proxy data.

4. Outlook

Enormous progress has been made since the climatic changes of the middle Miocene were first identified using benthic foraminifera geochemistry (Douglas & Savin, 1975; Shackleton & Kennett, 1975; Vincent & Berger, 1985; Woodruff et al., 1981). Nonetheless, further work is required, and our understanding of Miocene climate and the relationship between the spatial pattern of surface temperature, pCO_2 , and Antarctic ice volume is far from complete. Foremost amongst the outstanding questions is how the overall very positive $\delta^{18}O_{sw}$ values throughout the interval can be explained. Even at the maximum extent of the 95% confidence intervals, the new data from Modestou et al. (2020) would require an ice sheet similar in size to modern throughout the peak warmth of the MCO (400–800 ppm CO₂), whereas a recent ice sheet model suggest a greatly reduced Miocene ice volume at 2 × preindustrial CO₂ (Gasson et al., 2016). Future work specific to this aspect of Miocene climate should therefore focus on: understanding whether such large ice volumes can be reconciled with pCO₂ higher than present-day, understanding whether any unidentified bias exists in the benthic foraminifera δ^{18} O record, producing more high-latitude SST data, and of course, a renewed model and model-data intercomparison effort (the latter will be tackled by the forthcoming Miocene model



intercomparison project, MioMIP; Burls et al. (2021)). Taking a longer view, the benthic clumped isotope data presented by Modestou et al. (2020) are a first step toward producing a clumped isotope version of the Cenozoic benthic δ^{18} O stack – a mainstay in paleoclimatology which has been successively refined over the last two decades (Cramer et al., 2009; Westerhold et al., 2020; Zachos et al., 2001, 2008). Unraveling the full history of deep ocean temperature and ice volume over the last 65 Myr using this new approach will be a major achievement.

Despite the challenges that remain, new methodologies such as that utilized by Modestou et al. (2020) as well as an improved understanding of existing techniques mean that the confidence with which we interpret proxy data is changing. The paleoclimate literature is scattered with paradoxes, famous examples of which are the Eocene cool tropics paradox (see e.g. Barron, 1987; Pearson et al., 2001) and the related equable climate paradox (Huber & Caballero, 2011; Wing, 1994). The reasons for terming these phenomena as paradoxes are varied, but – in my view – derive in large part from our understanding of the proxies. That is not to imply that little is left to be understood about the climate system, rather that we, as a community, now have a proxy toolbox increasingly routed in a firm mechanistic understanding, while multiproxy approaches enable multiple aspects of the system of interest to be simultaneously constrained. Coupled with advances in climate and ice sheet modeling, and community model-data comparison projects (e.g. Hollis et al., 2019), I anticipate that the questions we ask will increasingly shift away from whether a past climate state was characterized by a given feature, toward how such features can be explained; the paradoxes will become increasingly rare. There has never been a more exciting or more important time to perform paleoclimate research.

Data Availability Statement

No new datasets are presented here. The derivation of Figure 1 is described in Text S1 and S2. The data shown in Figure 2 are available from Super et al. (2018), Sosdian et al. (2018), Shevenell et al. (2004), and Modestou et al. (2020).

Acknowledgments

The author would like to thank an anonymous reviewer and the editor for providing feedback that substantially improved this contribution, and Will Gray for discussion of benthic foraminifera geochemistry. The author declares no conflict of interest. Open Access funding enabled and organized by Projekt DEAL.

References

Adkins, J. F., Mcintyre, K., & Schrag, D. P. (2002). The Salinity, Temperature, and δ^{18} O of the Glacial Deep Ocean. *Science*, 298, 1769–1773. https://doi.org/10.1126/science.1076252

Affek, H. P. (2012). Clumped isotope paleothermometry: Principles, applications, and challenges. *Paleontological Society Papers*, 18(c), 101–114. https://doi.org/10.1017/S1089332600002576

Barron, E. J. (1987). Eocene equator-to-pole surface ocean temperature: A significant climate problem? *Paleoceanography*, 2(6), 729–739. https://doi.org/10.1029/PA002i006p00729

Burls, N. J., Bradshaw, C., De Boer, A. M., Herold, N., Huber, M., Pound, M., et al. (2021). Simulating Miocene warmth: insights from an opportunistic Multi-Model ensemble (MioMIP1). Earth and Space Science Open Archive. https://doi.org/10.1002/essoar.10505870.1

Chow, J. M., & Bart, P. J. (2003). West Antarctic Ice Sheet grounding events on the Ross Sea outer continental shelf during the middle Miocene. Palaeogeography, Palaeoclimatology, Palaeoecology, 198(1–2), 169–186. https://doi.org/10.1016/S0031-0182(03)00400-0

Cramer, B. S., Toggweiler, J. R., Wright, J. D., Katz, M. E., & Miller, K. G. (2009). Ocean overturning since the late cretaceous: Inferences from a new benthic foraminiferal isotope compilation. *Paleoceanography*, 24(4), 1–14. https://doi.org/10.1029/2008PA001683

Douglas, R. G., & Savin, S. M. (1975). Oxygen and carbon isotope analyses of Tertiary and Cretaceous microfossils from Shatsky Rise and other sites in the North Pacific Ocean. *Initial Reports of the Deep Sea Drilling Project*, *32*, 509–520. https://doi.org/10.1016/0377-8398(78)90004-X

Elderfield, H., Ferretti, P., Greaves, M., Crowhurst, S., McCave, I. N., Hodell, D., et al. (2012). Evolution of Ocean Temperature and ice volume through the Mid-Pleistocene climate transition. *Science*, *337*, 704–709. https://doi.org/10.1594/PANGAEA.786205

Evans, D., Badger, M. P. S., Zachos, J. C., Foster, G. L., Henehan, M. J., & Lear, C. H. (2018). No substantial long-term bias in the Cenozoic benthic foraminifera oxygen-isotope record. *Nature Communications*, 9, 17–19. https://doi.org/10.1038/s41467-018-05303-4

- Evans, D., & Müller, W. (2012). Deep time foraminifera Mg/Ca paleothermometry: Nonlinear correction for secular change in seawater Mg/Ca. *Paleoceanography*, 27(4), PA4205. https://doi.org/10.1029/2012PA002315
- Flower, B. P., & Kennett, J. P. (1995). Middle Miocene deepwater paleoceanography in the southwest Pacific: Relations with East Antarctic Ice Sheet development. *Paleoceanography*, *10*(6), 1095–1112. https://doi.org/10.1029/95PA02022
- Gasson, E., DeConto, R. M., Pollard, D., & Levy, R. H. (2016). Dynamic Antarctic ice sheet during the early to mid-Miocene. Proceedings of the National Academy of Sciences of the United States of America, 113(13), 3459–3464. https://doi.org/10.1073/pnas.1516130113

Ghosh, P., Adkins, J., Affek, H., Balta, B., Guo, W., Schauble, E., et al. (2006). ¹³C-¹⁸O bonds in carbonate minerals: A new kind of paleothermometer. *Geochimica et Cosmochimica Acta*, 70(6), 1439–1456. https://doi.org/10.1016/j.gca.2005.11.014

- Goldner, A., Herold, N., & Huber, M. (2014). Antarctic glaciation caused ocean circulation changes at the Eocene-Oligocene transition. *Nature*, 511(7511), 574–577. https://doi.org/10.1038/nature13597
- Gray, W. R., & Evans, D. (2019). Nonthermal influences on Mg/Ca in planktonic foraminifera: a review of culture studies and application to the Last Glacial Maximum. *Paleoceanography and Paleoclimatology*, *34*, 306–315. https://doi.org/10.1029/2018PA003517

Hamon, N., Sepulchre, P., Lefebvre, V., & Ramstein, G. (2013). The role of Eastern Tethys seaway closure in the middle Miocene climatic transition (ca. 14 Ma). *Climate of the Past*, 9(6), 2687–2702. https://doi.org/10.5194/cp-9-2687-2013



- Hansen, J., Sato, M., Russell, G., & Kharecha, P. (2013). Climate sensitivity, sea level and atmospheric carbon dioxide. *Philosophical Transactions of the Royal Society A: Mathematical, Physical & Engineering Sciences*, 371(2001), 20120294. https://doi.org/10.1098/rsta.2012.0294
 Holbourn, A., Kuhnt, W., Lyle, M., Schneider, L., Romero, O., & Andersen, N. (2014). Middle Miocene climate cooling linked to intensifi cation of eastern equatorial Pacifi c upwelling. *Geology*, 42(1), 19–22. https://doi.org/10.1130/G34890.1
- Holbourn, A., Kuhnt, W., Schulz, M., & Erlenkeuser, H. (2005). Impacts of orbital forcing and atmospheric carbon dioxide on Miocene ice-sheet expansion. *Nature*, 438(7067), 483–487. https://doi.org/10.1038/nature04123
- Holbourn, A., Kuhnt, W., Simo, J. A. T., & Li, Q. (2004). Middle Miocene isotope stratigraphy and paleoceanographic evolution of the northwest and southwest Australian margins (Wombat Plateau and Great Australian Bight). Palaeogeography, Palaeoclimatology, Palaeoceology, 208(1–2), 1–22. https://doi.org/10.1016/j.palaeo.2004.02.003
- Hollis, C. J., Jones, T. D., Anagnostou, E., Bijl, P. K., Cramwinckel, M. J., Cui, Y., et al. (2019). The DeepMIP contribution to PMIP4 : Methodologies for selection, compilation and analysis of latest paleocene and early Eocene climate proxy data, incorporating version 0.1 of the DeepMIP database. *Geoscientific Model Development*, 12(7), 3149–3206, https://doi.org/10.5194/gmd-12-3149-2019
- Horita, J., Zimmermann, H., & Holland, H. D. (2002). Chemical evolution of seawater during the Phanerozoic. Geochimica et Cosmochimica Acta, 66(21), 3733–3756. https://doi.org/10.1016/S0016-7037(01)00884-5
- Huber, M., & Caballero, R. (2011). The early Eocene equable climate problem revisited. *Climate of the Past*, 7(2), 603–633. https://doi.org/10.5194/cp-7-603-2011
- Inglis, G. N., Bragg, F., Burls, N. J., Cramwinckel, M. J., Evans, D., Foster, G. L., et al. (2020). Global mean surface temperature and climate sensitivity of the early Eocene Climatic Optimum (EECO), Paleocene-Eocene Thermal Maximum (PETM), and latest Paleocene. *Climate of the Past*, 16(5), 1953–1968. https://doi.org/10.5194/cp-16-1953-2020
- Knorr, G., & Lohmann, G. (2014). Climate warming during antarctic ice sheet expansion at the middle miocene transition. Nature Geoscience, 7(5), 376–381. https://doi.org/10.1038/ngeo2119
- Kozdon, R., Kelly, D. C., Kitajima, K., Strickland, A., Fournelle, J. H., & Valley, J. W. (2013). In situ δ¹⁸O and Mg/Ca analyses of diagenetic and planktic foraminiferal calcite preserved in a deep-sea record of the Paleocene-Eocene thermal maximum. *Paleoceanography*, 28, 517–528. https://doi.org/10.1002/palo.20048
- Langebroek, P. M., Paul, A., & Schulz, M. (2010). Simulating the sea level imprint on marine oxygen isotope records during the middle Miocene using an ice sheet-climate model. *Paleoceanography*, 25(4), 1–12. https://doi.org/10.1029/2008PA001704
- Lear, C. H., Coxall, H. K., Foster, G. L., Lunt, D. J., Mawbey, E. M., Rosenthal, Y., et al. (2015). Neogene ice volume and ocean temperatures: Insights from infaunal foraminiferal Mg/Ca paleothermometry. *Paleoceanography*, 30, 1437–1454. https://doi.org/10.1002/2015PA002833 Lear, C. H., Elderfield, H., & Wilson, P. A. (2000). Cenozoic deep-sea temperatures and global ice volumes from Mg/Ca in benthic fo-
- raminiferal calcite. *Science*, 287(5451), 269–272. https://doi.org/10.1126/science.287.5451.269 Lear, C. H., Mawbey, E. M., & Rosenthal, Y. (2010). Cenozoic benthic foraminiferal Mg/Ca and Li/Ca records: Toward unlocking tempera-
- tures and saturation states. *Paleoceanography*, 25(4), 1–11. https://doi.org/10.1029/2009PA001880
- Leutert, T. J., Sexton, P. F., Tripati, A., Piasecki, A., Ling, S., & Meckler, A. N. (2019). Sensitivity of clumped isotope temperatures in fossil benthic and planktic foraminifera to diagenetic alteration. *Geochimica et Cosmochimica Acta*, 257, 354–372. https://doi.org/10.1016/j. gca.2019.05.005
- Levy, R., Harwood, D., Florindo, F., Sangiorgi, F., Tripati, R., von Eynatten, H., et al. (2016). Antarctic ice sheet sensitivity to atmospheric CO₂ variations in the early to mid-Miocene. Proceedings of the National Academy of Sciences of the United States of America, 113(13), 3453–3458. https://doi.org/10.1073/pnas.1516030113
- Lisiecki, L. E., & Raymo, M. E. (2005). A Pliocene-Pleistocene stack of 57 globally distributed benthic δ¹⁸O records. *Paleoceanography*, 20(1), PA1003. https://doi.org/10.1029/2004PA001071
- Marchitto, T. M., Curry, W. B., Lynch-Stieglitz, J., Bryan, S. P., Cobb, K. M., & Lund, D. C. (2014). Improved oxygen isotope temperature calibrations for cosmopolitan benthic foraminifera. Geochimica et Cosmochimica Acta, 130, 1–11. https://doi.org/10.1016/j.gca.2013.12.034
- Meckler, A N., Ziegler, M., Millán, M. I., Breitenbach, S. F. M., & Bernasconi, S. M. (2014). Long-term performance of the Kiel carbonate device with a new correction scheme for clumped isotope measurements. *Rapid Communications in Mass Spectrometry*, 28(15), 1705– 1715. https://doi.org/10.1002/rcm.6949
- Meinicke, N., Ho, S. L., Hannisdal, B., Nürnberg, D., Tripati, A., Schiebel, R., et al. (2020). A robust calibration of the clumped isotopes to temperature relationship for foraminifers. *Geochimica et Cosmochimica Acta*, 270, 160–183. https://doi.org/10.1016/j.gca.2019.11.022
- Miller, K. G., Browning, J. V., John Schmelz, W., Kopp, R. E., Mountain, G. S., & Wright, J. D. (2020). Cenozoic sea-level and cryospheric evolution from deep-sea geochemical and continental margin records. *Science Advances*, 6(20), eaaz1346. https://doi.org/10.1126/sciadv.aaz1346
- Modestou, S. E., Leutert, T. J., Fernandez, A., Lear, C. H., & Meckler, A. N. (2020). Warm middle Miocene Indian Ocean bottom water temperatures: comparison of clumped isotope and Mg/Ca based estimates. *Paleoceanography and Paleoclimatology*, 35(11), e2020PA003927. https://doi.org/10.1029/2020PA003927
- O'Brien, C. L., Huber, M., Thomas, E., Pagani, M., Super, J. R., Elder, L. E., et al. (2020). The enigma of Oligocene climate and global surface temperature evolution. *Proceedings of the National Academy of Sciences of the United States of America*, 117(41), 25302–25309. https:// doi.org/10.1073/pnas.2003914117
- Pearson, P. N., Ditchfield, P. W., Singano, J., Harcourt-Brown, K. G., Nicholas, C. J., Olsson, R. K., et al. (2001). Warm tropical sea surface temperatures in the Late Cretaceous and Eocene epochs. *Nature*, 413(6855), 481–487. https://doi.org/10.1038/35097000
- Pollard, D., & Deconto, R. M. (2016). Contribution of Antarctica to past and future sea-level rise. Nature, 531(7596):591–597. https://doi. org/10.1038/nature17145
- Rosenthal, Y., Boyle, E. A., & Slowey, N. (1997). Temperature control on the incorporation of magnesium, strontium, fluorine, and cadmium into benthic foraminiferal shells from Little Bahama Bank: Prospects for thermocline paleoceanography. *Geochimica et Cosmochimica Acta*, 61(17), 3633–3643. https://doi.org/10.1016/S0016-7037(97)00181-6
- Savin, S. M. (1977). The History of the Earth's Surface Temperature During the Past 100 Million Years. Annual Review of Earth and Planetary Sciences, 5, 319–355. https://doi.org/10.1146/annurev.ea.05.050177.001535
- Schmid, T. W., & Bernasconi, S. M. (2010). An automated method for 'clumped-isotope' measurements on small carbonate samples. Rapid Communications in Mass Spectrometry, 24, 1955–1963. https://doi.org/10.1002/rcm
- Shackleton, N. J., & Kennett, J. P. (1975). Paleotemperature hisotry of the Cenozoic and the initiation of Antarctic glaciation: Oxygen and carbon isotope analyses in DSDP Sites 277, 279, and 281. *Initial Reports of the Deep Sea Drilling Project*, 29, 743–755. https://doi. org/10.2973/DSDP.PROC.29.117.1975
- Shevenell, A. E., Kennett, J. P., & Lea, D. W. (2004). Middle Miocene Southern Ocean Cooling and Antarctic Cryosphere Expansion. Science, 305, 1766–1770. https://doi.org/10.1126/science.1100061



- Smart, C. W., Thomas, E., & Ramsay, A. T. S. (2007). Middle-late Miocene benthic foraminifera in a western equatorial Indian Ocean depth transect: Paleoceanographic implications. *Palaeogeography, Palaeoclimatology, Palaeoecology, 247*(3–4), 402–420. https://doi. org/10.1016/j.palaeo.2006.11.003
- Sosdian, S. M., Greenop, R., Hain, M. P., Foster, G. L., Pearson, P. N., & Lear, C. H. (2018). Constraining the evolution of Neogene ocean carbonate chemistry using the boron isotope pH proxy. *Earth and Planetary Science Letters*, 498, 362–376. https://doi.org/10.1016/j. epsl.2018.06.017
- Stap, L. B., Sutter, J., Knorr, G., Stärz, M., & Lohmann, G. (2019). Transient Variability of the Miocene Antarctic Ice Sheet Smaller Than Equilibrium Differences. *Geophysical Research Letters*, 46(8), 4288–4298. https://doi.org/10.1029/2019GL082163
- Stap, L. B., WalVan De, R. S. W., Boer, B. D., Bintanja, R., & Lourens, L. J. (2016). The MMCO-EOT conundrum: Same benthic δ^{18} O, different CO₂. *Paleoceanography and Paleoclimatology*, 31(9), 1270–1282. https://doi.org/10.1002/2016PA002958
- Steinthorsdottir, M., Coxall, H. K., de Boer, A. M., de Huber, M., Barbolini, N., Bradshaw, C. D., et al. (2020). The Miocene: The future of the past. *Paleoceanography and paleoclimatology*, e2020PA004037. https://doi.org/10.1029/2020PA004037
- Super, J. R., Thomas, E., Pagani, M., Huber, M., O'Brien, C., & Hull, P. M. (2018). North Atlantic temperature and pCO₂ coupling in the early-middle Miocene. *Geology*, 46(6), 519–522. https://doi.org/10.1130/G40228.1
- Tierney, J. E., Poulsen, C. J., Montañez, I. P., Bhattacharya, T., Feng, R., Ford, H. L., et al. (2020). Review summary past climates inform our future. *Science*, 370(6517), eaay3701. https://doi.org/10.1126/science.aay3701
- Tripati, A. K., Hill, P. S., Eagle, R. A., Mosenfelder, J. L., Tang, J., Schauble, E. A., et al. (2015). Beyond temperature: Clumped isotope signatures in dissolved inorganic carbon species and the influence of solution chemistry on carbonate mineral composition. *Geochimica et Cosmochimica Acta*, 166, 344–371. https://doi.org/10.1016/j.gca.2015.06.021
- Vincent, E., & Berger, W. H. (1985). Carbon dioxide and polar cooling in the Miocene: The Monterey hypothesis. In The Carbon Cycle and Atmospheric CO₂: Natural Variations Archean to Present. *Geophysical Monograph Series* (Vol. 32, pp. 455–468). https://doi.org/10.1029/ GM032p0455
- Warren, B. A. (1981). Transindian hydrographic section at Lat. 18°S: Property distributions and circulation in the South Indian Ocean. Deep-Sea Research, Part A: Oceanographic Research Papers, 28A(8), 759–788. https://doi.org/10.1016/s0198-0149(81)80001-5
- Westerhold, T., Marwan, N., Drury, A. J., Liebrand, D., Agnini, C., Anagnostou, E., et al. (2020). An astronomically dated record of Earth's climate and its predictability over the last 66 million years. *Science*, 1387, 1383–1387. https://doi.org/10.1126/science.aba6853
- Wilson, D. S., Pollard, D., Deconto, R. M., Jamieson, S. S. R., & Luyendyk, B. P. (2013). Initiation of the West Antarctic Ice Sheet and estimates of total Antarctic ice volume in the earliest Oligocene. *Geophysical Research Letters*, 40(16), 4305–4309. https://doi.org/10.1002/ grl.50797
- Wing, S. L. (1994). Paleoclimate, Proxies, Paradoxes, and Predictions. Palaios, 9(2), 121-123. https://doi.org/10.2307/3515100
- Woo, M., & Pattiaratchi, C. (2008). Hydrography and water masses off the western Australian coast. Deep-Sea Research Part I Oceanographic Research Papers, 55(9), 1090–1104. https://doi.org/10.1016/j.dsr.2008.05.005
- Woodruff, F., & Savin, S. M. (1989). Miocene deepwater oceanography. Paleoceanography, 4(1), 87-140. https://doi.org/10.1029/ PA004i001p00087
- Woodruff, F., & Savin, S. M. (1991). Mid-Miocene isotope stratography in the deep sea: High-resolution correlations, paleoclimate cycles, and sediment preservation. *Paleoceanography*, 6(6), 755–806. https://doi.org/10.1029/91PA02561
- Woodruff, F., Savin, S. M., & Douglas, R. G. (1981). Miocene stable isotope record: A detailed deep Pacific Ocean study and its paleoclimatic implications. Science, 212(4495), 665–668. https://doi.org/10.1126/science.212.4495.665
- Zachos, J. C., Dickens, G. R., & Zeebe, R. E. (2008). An early Cenozoic perspective on greenhouse warming and carbon-cycle dynamics. *Nature*, 451(7176), 279–283. https://doi.org/10.1038/nature06588
- Zachos, J. C., Pagani, M., Sloan, L., Thomas, E., & Billups, K. (2001). Trends, rhythms, and aberrations in global climate 65 Ma to present. *Science*, 292, 686–693. https://doi.org/10.1126/science.1059412
- Zeebe, R. E. (1999). An explanation of the effect of seawater carbonate concentration on foraminiferal oxygen isotopes. *Geochimica et Cosmochimica Acta*, 63(13–14), 2001–2007.